\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{}{}{}
 
%%%% PACKAGES%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%  
\usepackage{tikz} 
%\usepackage{a4wide}
%\usepackage[notcite,notref,color]{showkeys} 

\usepackage{amsmath,amssymb,amsthm,bbm} 
\usepackage{enumitem} 
%\usepackage{enumerate}
\usepackage[overload]{textcase}
 
\usepackage{epsfig,psfrag,color,float,graphics} 
\usepackage{verbatim}

\usepackage[colorlinks,bookmarks,linkcolor=black,citecolor=black]{hyperref} 
\usepackage{bookmark}
\usepackage{amsthm}

\theoremstyle{plain}


%%%% LISTS %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
 
\setlist{itemsep=5pt,parsep=0pt,topsep=4pt,partopsep=0pt,leftmargin=1.5\parindent} 

%%% LABELS 
 
\def\itm#1{\rm ({#1})} 
\def\itmit#1{\itm{\it #1\,}} 
\def\rom{\itmit{\roman{*}}} 
\def\abc{\itmit{\alph{*}}}
\def\abcstar{\itmit{\alph{*}*}}
\def\arab{\itmit{\arabic{*}}} 
\def\endofClaim{\hfill\scalebox{.6}{$\Box$}}

%%%% DEFLETS %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
 
\let\subset\subseteq  
\let\eps\varepsilon 
\let\rho\varrho 
%\def\dcup{\buildrel\textstyle.\over\cup}
% \def\dcup{\dot\cup}  
\def\dcup{\cup}

\def\subsc#1{\textsc{\MakeTextLowercase{#1}}} % needs package textcase

%%% CALLIGRAPHICS

\def\cB{\mathcal{B}}
\def\cC{\mathcal{C}}
\def\cD{\mathcal{D}}
\def\cF{\mathcal{F}}
\def\cG{\mathcal{G}}
\def\cH{\mathcal{H}}
\def\cK{\mathcal{K}}
\def\cM{\mathcal{M}}
\def\cP{\mathcal{P}}
\def\cR{\mathcal{R}}
\def\cS{\mathcal{S}}
\def\cT{\mathcal{T}}
\def\cW{\mathcal{W}}
\def\cY{\mathcal{Y}}

\def\bfu{\mathbf{u}}

\def\le{\leqslant}
\def\ge{\geqslant}
\def\leq{\leqslant}
\def\geq{\geqslant}


%%%% THEOREMS %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
 
 \newtheorem*{theorem*}{Theorem}
 
 
\newtheorem{theorem}{Theorem}%[section] 
\newtheorem*{SzRL}{Szemer\'edi's Regularity Lemma}
\newtheorem*{SzRLdeg}{Szemer\'edi's Regularity Lemma (minimum degree form)}
\newtheorem*{Count}{Counting Lemma}
\newtheorem{lemma}[theorem] {Lemma}    
\newtheorem{corollary}[theorem]{Corollary}    
\newtheorem{prop}[theorem] {Proposition}   
\newtheorem{fact}[theorem]{Fact} 
\newtheorem{obs}[theorem] {Observation}  
\newtheorem{conj}[theorem]{Conjecture}
\newtheorem{prob}[theorem]{Problem} 
\newtheorem{property}{Property}
\newtheorem{qu}[theorem]{Question}  
\newtheorem{claim}[theorem]{Claim} 
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{remark}[theorem]{Remark}
\theoremstyle{remark} 
 

\newcommand{\oldqed}{}

\newenvironment{claimproof}[1][Proof]{
  \renewcommand{\oldqed}{\qedsymbol}
  \renewcommand{\qedsymbol}{\endofClaim}
  \begin{proof}[#1]
}{
  \end{proof}
  \renewcommand{\qedsymbol}{\oldqed}
}
%%%% COMMANDS %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
 
\newcommand{\redcomment}[1]{\marginpar{\baselineskip=.6\baselineskip 
    \textcolor{red}{\sffamily\footnotesize #1}}} 
% \newcommand{\comment}[1]{} 
 
% explanations above <,>,=,\le,\ge 
\newcommand{\By}[2]{\overset{\mbox{\tiny{#1}}}{#2}} 
\newcommand{\ByRef}[2]{   \By{\eqref{#1}}{#2} } 
\newcommand{\eqBy}[1]{    \By{#1}{=} } 
\newcommand{\lBy}[1]{     \By{#1}{<} } 
\newcommand{\gBy}[1]{     \By{#1}{>} } 
\newcommand{\leBy}[1]{    \By{#1}{\le} } 
\newcommand{\geBy}[1]{    \By{#1}{\ge} } 
\newcommand{\eqByRef}[1]{ \ByRef{#1}{=} } 
\newcommand{\lByRef}[1]{  \ByRef{#1}{<} } 
\newcommand{\gByRef}[1]{  \ByRef{#1}{>} } 
\newcommand{\leByRef}[1]{ \ByRef{#1}{\le} } 
\newcommand{\geByRef}[1]{ \ByRef{#1}{\ge} } 
 
\newcommand{\REALS}{\mathbb{R}} 
\newcommand{\RATS}{\mathbb{Q}} 
\newcommand{\NATS}{\mathbb{N}} 
\newcommand{\bigO}{\mathcal{O}}
\newcommand{\Prob}{\mathbb{P}}
\newcommand{\Exp}{\mathbb{E}}


\def\AA{\mathbb{A}}
\def\Ex{\mathbb{E}}
\def\FF{\mathbb{F}}
\def\Pr{\mathbb{P}}
\def\QQ{\mathbb{Q}}
\def\RR{\mathbb{R}}
\def\ZZ{\mathbb{Z}}
\def\MM{\mathbb{M}}
\def\NN{\mathbb{N}}


\newcommand{\ds}{\displaystyle}
\newcommand{\ul}{\underline}
\newcommand{\ol}{\overline}
\newcommand{\girth}{\textup{girth}}
\newcommand{\Kn}{\textup{Kn}}
\newcommand{\Bor}{\textup{Bor}}
\newcommand{\BH}{\textup{BH}}
\newcommand{\Haj}{\textup{Haj}}
\newcommand{\Bin}{\textup{Bin}}
\newcommand{\Aut}{\textup{Aut}}

\newcommand{\EMAIL}[1]{  \textit{E-mail}: \texttt{#1} } 

\newcommand{\tpl}[2]{\mathbf{#1}^{#2}}
\newcommand{\nbh}{N}
\newcommand{\ang}{\measuredangle}

\newcommand{\sat}{\mathrm{sat}}
\newcommand{\satp}{\mathrm{sat_p}}
\newcommand{\exsatp}{\mathrm{exsat_p}}
\newcommand{\vbad}{\mathrm{V_{bad}}}

\DeclareMathOperator{\ex}{ex}
\usetikzlibrary{calc, shapes, backgrounds}
\usetikzlibrary{arrows,decorations.pathmorphing,positioning,fit,petri}

%opening
\title{Ramsey Numbers of Connected Clique Matchings}
\author{Barnaby Roberts\\
\small Department of Mathematics\\[-0.8ex] 
\small London School of Economics\\[-0.8ex] 
\small Houghton Street, London WC2A 2AE, U.K.\\
\small\texttt{b.j.roberts@lse.ac.uk}
}
\date{\dateline{Jul 6, 2016}{Feb 2, 2017}{}}
%\keywords{Ramsey Theory, Connected Graphs, Clique Matchings}
\begin{document}
\begin{thebibliography}{10}

\bibitem{AllBriSko}
P.~Allen, G.~Brightwell, and J.~Skokan.
\newblock Ramsey goodness and otherwise.
\newblock {\em Combinatorica}, 33:125--160, 2013.

\bibitem{AllSkoRob}
P.~Allen, J.~Skokan, and B.~Roberts.
\newblock The ramsey numbers of squares of paths and cycles.
\newblock {\em In Preparation}.

\bibitem{Burr}
S.~Burr.
\newblock Ramsey numbers involving graphs with long suspended paths.
\newblock {\em Journal of the London Mathematical Society}, 2:405--413,
  1981.

\bibitem{BurrErdSpen}
S.~Burr, P.~Erd{\H{o}}s, and J.~Spencer.
\newblock Ramsey theorems for multiple copies of graphs.
\newblock {\em Transactions of the American Mathematical Society}, 209:87--99,
  1975.

\bibitem{ConUB}
D.~Conlon.
\newblock A new upper bound for diagonal ramsey numbers.
\newblock {\em Annals of Mathematics}, 170:941--960, 2009.

\bibitem{ErdLB}
P.~Erd{\H{o}}s.
\newblock Some remarks on the theory of graphs.
\newblock {\em Bulletin of the American Mathematical Society}, 53:292--294,
  1947.

\bibitem{ErdSzek}
P.~Erd{\H{o}}s and G.~Szekeres.
\newblock A combinatorial problem in geometry.
\newblock {\em Compositio Mathematica}, 2:463--470, 1935.

\bibitem{GraRodRuc}
R.~Graham, V.~R{\"o}dl, and A.~Rucinski.
\newblock On graphs with linear ramsey numbers.
\newblock {\em Journal of Graph Theory}, 35:176--192, 2000.

\bibitem{GyaSark}
A.~Gy\'arf\'as and G.~N. S\'ark\"{o}zy.
\newblock Ramsey number of a connected triangle matching.
\newblock {\em arXiv preprint arXiv:1509.05530}, 2015.

\bibitem{SpenLB}
J.~Spencer.
\newblock Ramsey's theorem - a new lower bound.
\newblock {\em Journal of Combinatorial Theory, Series A}, 18:108--115,
  1975.

\end{thebibliography}



\maketitle

\begin{abstract}
We determine the Ramsey number of a connected clique matching. That is, we show that if $G$ is a $2$-edge-coloured 
complete graph on $(r^2-r-1)n-r+1$ vertices, then there is a 
monochromatic connected subgraph containing $n$ disjoint copies of 
$K_r$ for $n$ sufficiently large.  This number of vertices cannot be reduced.
\end{abstract}

\section{Introduction}
For a graph $G$, the Ramsey number $R(G)$ is defined to be the smallest integer $N$ such that every 2-colouring of the edges of the complete graph on $N$ vertices contains a monochromatic subgraph isomorphic to $G$.
The most fundamental problem in Ramsey theory is determining the order of magnitude of the Ramsey numbers of cliques.
An exponential upper bound was given by Erd\H{o}s and Szekeres \cite{ErdSzek}, and in an early use of the probabilistic method Erd\H{o}s \cite{ErdLB} gave an exponential lower bound.
In spite of progress being made on both upper and lower bounds (see \cite{ConUB,SpenLB}) even the size of the exponent is not known asymptotically.
It becomes easier however to look for multiple copies of the cliques.

The Ramsey numbers of multiple copies of graphs were studied in \cite{BurrErdSpen} by Burr, Erd\H{o}s, and Spencer who determined the Ramsey number of $nK_3$ exactly and of multiple copies of a general graph $G$ up to a constant depending only on $G$.
In particular they showed $R(nK_3)=5n$ and for $r\geq 4$ that $(2r-1)n-1 \leq R(nK_r)\leq (2r-1)n +C_r$, determining the Ramsey number of a $K_r$-matching up to a constant.

The aim of this note is to add a connectivity requirement, studying the Ramsey numbers of connected copies of cliques.
Although not technically a Ramsey number by the definition given above, we let $R(c(nH))$ denote the least $N$ such that every $2$-colouring of the edges of $K_N$ contains a monochromatic copy of $nH$ in a connected component of the same colour.
This was first studied by Gy\'arf\'as and S\'ark\"{o}zy \cite{GyaSark} who solved the problem for $H=K_3$ showing $R(c(nK_3))=7n-2$.
We solve the problem for all larger cliques.

\begin{theorem}\label{thm:main}
For $r\geq 4$ and $n \geq R(K_r)$ we have \[R(c(nK_r)) = (r^2 -r +1)n -r +1\,. \]
\end{theorem}


The connectivity requirement here proves to be very significant as for large values of $r$ the Ramsey number of a connected $K_r$-matching is approximately $r/2$ times larger than that of a standard $K_r$-matching.



The lower bound in Theorem~\ref{thm:main} is a special case of the following observation of Burr~\cite{Burr}.
\begin{prop}\label{prop:Burr}
For every connected graph $G$ containing at least one edge
\begin{equation*}
R(G) \geq (\chi(G)-1)(|G|-1) + \sigma(G)
\end{equation*}
where $\sigma(G)$ is the smallest size of a colour class over all $\chi(G)$-proper-colourings of $G$.
\end{prop}
To see this partition the vertices of $K_N$, with $N=(\chi(G)-1)(|G|-1) + \sigma(G)-1$, into parts $X_1,\ldots,X_{\chi(G)-1}$ and $Y$ such that $|Y|=\sigma(G)-1$ and $|X_i|=|G|-1$ for each $i$, as in Figure~\ref{fig}.
Colour all edges within each part with blue and colour all edges between different parts with red.
There cannot be a blue copy of $G$ since all connected components of blue are too small.
There cannot be a copy of $G$ in red since the red edges form a $\chi(G)$-partite graph where the smallest part is too small.


% PICTURE of BURR CONSTRUCTION
%\begin{minipage}{61mm}
\begin{figure}[H]
\[%begin{center}
\begin{tikzpicture}[line width=1pt,every node/.style={circle,fill,draw,inner sep=0,minimum size=4pt}]
\node (A) at ( 1,1.25) [circle, fill=blue!70,inner sep=8pt,label=left:$|G|-1$]{};
\node (B) at ( 1,-1.25) [circle, fill=blue!70,inner sep=8pt,label=left:$|G|-1$]{};
\node (C) at ( 4,2.5) [circle, fill=blue!70,inner sep=8pt,label=right:$|G|-1$]{};
\node (D) at ( 4,-2.5) [circle, fill=blue!70,inner sep=8pt,label=right:$|G|-1$]{};
\node (E) at ( 6,0) [circle, fill=blue!80,inner sep=5pt,label=right:$|\sigma(G)|-1$]{};
  \foreach \from/\to in {A/B,A/C,A/D,A/E,B/C,B/D,B/E,C/D,C/E,D/E}
  \draw [red!60,line width=5pt] (\from) -- (\to);
\end{tikzpicture}
\]%end{center}
\vskip-15pt
\caption{\label{fig}}
\end{figure}
%\end{minipage}

\smallskip
This construction led Burr to conjecture that for any $\Delta,k \in \NN$ there exists an $n_0$ such that any connected graph $G$ on $n \geq n_0$ vertices with chromatic number $k$ and maximum degree at most $\Delta$ satisfies $R(G)=(k-1)(n-1) +\sigma(G)$.
Burr's conjecture was proven to be false by Graham, R\"odl, and Rucinski~\cite{GraRodRuc}.
Our result shows that Burr's conjecture does hold in the case of connected copies of cliques.
We conjecture the following
\begin{conj}
For any graph $H$ there exists $n_0(H)$ such that for all $n \geq n_0$ we have
\[R(c(nH))=(\chi(H)-1)(n|H|-1)+n\sigma(H)\,.  \]
\end{conj}
The lower bound holds by Proposition~\ref{prop:Burr}.


There is another stronger connectivity requirement on disjoint cliques for which it is interesting to study the Ramsey numbers.
Suppose $P$ and $Q$ are copies of $K_r$.  We say they are $K_r$-connected if there exist copies $C_1,\ldots,C_t$ of $K_r$ such that in the sequence $P,C_1,\ldots,C_t,Q,$ consecutive copies of $K_r$ share $r-1$ vertices.
A construction of Allen, Brightwell, and Skokan in \cite{AllBriSko} shows that if we wish to find a monochromatic set of $n$ disjoint copies of $K_r$ which are $K_r$-connected in $K_N$ we need $N \geq nr^2 -2r +2$.
This construction was given as a lower bound for the Ramsey number $R(C_{rn}^{r-1})$ where $C_m^k$ is the $m$-th power of a cycle, obtained by joining all vertices of $C_m$ at distance at most $k$.
In \cite{AllSkoRob}, with Allen and Skokan we determine the Ramsey number $R(C^2_m)$ for large $m$ via studying the Ramsey numbers of $K_3$-connected copies of disjoint triangles.  The difference between that result and the result of this paper gives some intuition for when Burr's construction is best or not.  The requirement that cliques are just connected allows Burr's construction to be best however if we require a stronger connectivity condition then this is no longer the case.




\section{Proof of Theorem~\ref{thm:main}}
The lower bound for Theorem~\ref{thm:main} follows from the construction of Burr given in the previous section.
Before giving a proof of the upper bound we sketch the main ideas.

For $r\geq 4$ and $n\geq R(K_r)$, consider a $2$-colouring of $G=K_N$ with $N= (r^2 -r +1)n -r +1$.
Since in any $2$-colouring of a complete graph one of the colours is connected, we assume $G$ is connected in red and look for either a red copy of $nK_r$ or a blue connected copy of $nK_r$.
We then show that $G$ can be partitioned into a maximal set of disjoint red copies of $K_r$ and a set of $r-1$ large blue cliques such that between any two of the blue cliques all edges are red.
We call these blue cliques $B_1,\ldots,B_{r-1}$, and the union of their vertex sets $B$.
We let $\cR$ denote the maximum set of red copies of $K_r$, and we let $R$ denote the vertex set of $\cR$.
We then consider edges between the red copies of $K_r$ in $\cR$ and the sets $B_1,\ldots,B_{r-1}$.
We show that each clique of $\cR$ is of one of two types (see claim~\ref{cl:cliquetypes}) with regards to how the edges between that clique and $B$ are coloured.
Furthermore each type gives a way of assigning vertices of the red $K_r$ to some $B_i$ such that almost all of the edges between the assigned vertex and $B_i$ are blue.
For each $i=1,\ldots,r-1$, we let $D_i$ denote the union of $B_i$ along with the vertices of $R$ that were assigned to $B_i$.
We then use an averaging argument to show that there exists an $i$ such that $|D_i|\geq rn$ and we look for a blue $nK_r$ on this $D_i$.
Since $B_i$ was a blue clique and vertices assigned to $B_i$ were connected to $B_i$ in blue, this $nK_r$ is connected in blue.
If more than $(r-1)n$ of the vertices of $D_i$ came from $B_i$ we can find the $nK_r$ greedily.
Otherwise we use more information coming from the types of red $K_r$ vertices assigned to $B_i$ came from to find an $nK_r$.
There is one special case for which this method fails.



\begin{proof}[Proof of Theorem~\ref{thm:main}]
For $r\geq 4$ and $n\geq R(K_r)$, let $N= (r^2 -r +1)n -r +1$ and consider a red/blue edge colouring of $G=K_N$.
Any $2$-colouring of $K_N$ is connected in one of the colour, since if Blue has more than one connected component, Red contains a complete multi-partite graph between blue components which connects the graph in red.
Without loss of generality we assume $G$ is connected in red.
Consider a maximal set of vertex disjoint red copies of $K_r$.
We call this set of cliques $\cR$ and its vertex set $R$.
Note that $|R| \leq r(n-1)$, otherwise we would be done.
Greedily looking for vertex disjoint blue copies of $K_r$ on the rest of the graph results in covering all but at most $R(K_r)-1$ vertices of the graph with monochromatic copies of $K_r$.
Call this set of uncovered vertices $Z$.

Let $B$ denote the vertices covered by these blue cliques.
We next partition $B$ into blue connected components.

Between components all edges are red, so there cannot be more than $r-1$ such components or we would have a new red $K_r$ to add to $\cR$.
This would contradict the maximality of $\cR$.
There also cannot be fewer than $r-1$ such components, or, since each component is no larger than $r(n-1)$, we would have 
\[|B|+|R|+|Z| \leq (r-2)r(n-1)+ r(n-1) +R(K_r)-1 = (r^2 -r)n -r^2 +r -1 +R(K_r),\]
which is a contradiction, since for $n>R(K_r$ this is less than $N$.
Therefore there must be exactly $r-1$ blue components and we call them $B'_1,\ldots,B'_{r-1}$.
If any of these components contained a red edge we would have another red $K_r$ to add to $\cR$, and so each $B'_i$ is a blue clique.
Since the blue components are disconnected, each vertex in $Z$ can be adjacent in blue to vertices of at most one $B'_i$.
If any vertex in $Z$ had a red neighbour in each $B'_i$ we would have a new red $K_r$ to add to $\cR$.
Therefore all vertices in $Z$ are adjacent to $r-2$ of the $B'_i$ entirely in red and the remaining blue component entirely in blue.
For each $i=1,\ldots,r-1$, form $B_i$ by adding to $B'_i$ the vertices of $Z$ that are adjacent in blue to all of $B'_i$.
Note that there cannot be any red edges in any $B_i$ or we would have a new red $K_r$ to add to $\cR$ and also between any two distinct $B_i$ all edges are red since they are not blue connected.

We now consider the colour of edges between the cliques of $\cR$ and the sets $B_1,\ldots,B_{r-1}$.
First, recalling that all vertices in $R$ are adjacent in blue to vertices in at most one $B_i$, we say a vertex of $R$ is \emph{paired with $B_i$} if it is adjacent in red to all $B_j$ for all $j \neq i$.
The following claim identifies two possible properties of cliques of $\cR$.

\begin{claim}\label{cl:cliquetypes}
Let $C$ be a red $K_r$ from $\cR$.  Then one of the following holds:
\begin{enumerate}
\item[(i)]\label{cl:it:1} For each $B_i$ there is a vertex in $C$ adjacent in blue to all but at most one vertex of~$B_i$.
\item[(ii)]\label{cl:it:2} For all but two values of $i$ there is exactly one vertex of $C$ which is adjacent in blue to all vertices of $B_i$.  Furthermore there is a $j$ such that the three remaining vertices of $C$ are adjacent in blue to all of $B_j$.
\end{enumerate}



\end{claim}
\begin{claimproof}
Consider some $C \in \cR$.
Each vertex in $C$ can have blue neighbours in at most one $B_i$ and so is entirely adjacent in red to the remaining $r-2$ blue components.


Suppose firstly that some $c_i \in C$ is paired with $B_i$ but also has at least one neighbour in red in $B_i$.
Then if two other vertices $c_j,c'_j \in C$ were paired with the same $B_j$, we could break up $C$ to make two new red copies of $K_r$, contradicting the maximality of $\cR$.
One of these red cliques uses $c_i$, its red neighbour in $B_i$ and vertices of $B_k$ for $k \neq i$.
The other uses $c_j$, $c'_j$ and vertices of each $B_k$ for $k \neq j$.
Therefore, if such a $c_i \in C$ exists, all remaining vertices of $C$ are paired with distinct components $B_j$.
Furthermore, if any of them has more than one red neighbour in the set they are paired with, we could make two new red copies of $K_r$, again contradicting the maximality of $\cR$.
One of these would use $c_i$ as before, and the other would use the other vertex with red neighbours in the set it is paired with.
Thus, if $c_i$ as above exists, $(i)$ holds.

Secondly, suppose all vertices in $C$ are adjacent entirely in blue to the set they are paired with.
There cannot be two vertices $c_i, c'_i$ paired with some $B_i$ and another two vertices $c_j, c'_j$ paired with some $B_j$.
If there were, we would create two new red copies of $K_r$ using $c_i$, $c'_i$ and a vertex from each $B_k$ with $k \neq i$ for one, and $c_j$, $c'_j$ and a vertex from each $B_k$ with $k \neq j$ for the other.
This would again break the maximality of $\cR$.
There also cannot be four or more vertices paired with the same $B_i$, or we could use two pairs of them to make two new red copies of $K_r$ along with vertices from $B_k$ for $k \neq i$.
Therefore we either have some $B_i$ with two vertices of $C$ paired with it and every other $B_j$ has one vertex paired with them, as in $(i)$, or we have some $B_i$ with three vertices paired with it and all but one of the remaining components $B_j$ have one vertex paired with them, as in $(ii)$.
\end{claimproof}

The next claim tells us that if we have two sets of three vertices from distinct cliques of $\cR$ that are paired with the same $B_i$, then the edges between these two sets are all blue.
\begin{claim}\label{cl:bluebetween}
Let $C$ and $C'$ be cliques of $\cR$ with vertices $x_1,x_2,x_3 \in C$ and $y_1,y_2,y_3 \in C'$ all paired with the same $B_i$.
Then all the edges $x_jy_k$ for $j,k \in \{1,2,3\}$ are blue.
\end{claim}
\begin{claimproof}
Suppose for contradiction and without loss of generality that $x_1y_1$ is red.
Then we could create three new red copies of $K_r$ at the cost of $C$ and $C'$, contradicting the maximality of $\cR$.
These three copies of $K_r$ would use the pairs of vertices $\{x_1,y_1\}$, $\{x_2,x_3\}$ and $\{y_2,y_3\}$ along with vertices from $B_j$ for $j \neq i$.
\end{claimproof}




We now use Claim~\ref{cl:cliquetypes} to partition the red cliques of $\cR$ depending on which option of the claim they satisfy.
Let $S \subset \cR$ be the set of cliques satisfying $(i)$, and $T \subset \cR$ be the set of cliques satisfying $(ii)$.
For each $C \in S$ and each $B_i$ there is at least one vertex of $C$ which is adjacent in blue to all but at most one vertex of $B_i$.
For each $i$, construct $S_i$ by selecting one such vertex from each $C \in S$.
For each $C \in T$, all vertices are adjacent entirely in blue to exactly one $B_i$.
For each $i$, construct $T_i$ by taking the vertices of each clique of $T$ that are entirely adjacent in blue to $B_i$.
Let $D_i = B_i \cup S_i \cup T_i$.
We further split up $T_i$, into sets $T_i^*$ and $T_i^\Delta$, depending on whether one or three vertices from that red $K_r$ were added to $T_i$.
Given a vertex $u \in T_i$ let $C$ be the clique of $T$ containing $u$.
If $u$ is the only vertex of $C$ belonging to $T_i$ then $u$ belongs to $T_i^*$.
Otherwise three vertices of $C$ must belong to $T_i$ in which case all three belong to $T_i^\Delta$.
Observe that $|T_i|=|T_i^*|+|T_i^\Delta|$ and $n-|S_i|-|T_i^*|\geq \frac13 |T_i^\Delta|+1$.

We shall find a blue connected copy of $nK_r$ on a $D_i$ such that either $|D_i|\geq rn+1$, or $|D_i|\geq rn$ and $|T_i^\Delta| =0$.
We first proceed to show that such a $D_i$ exists.


Because $\bigcup_{i=1}^{r-1}D_i$ covers all vertices of the graph except for one from each clique of $S$, we have
\begin{equation*}
N=(r^2 -r +1)n-r+1=|S|+ \sum_{i=1}^{r-1} |D_i| \,.
\end{equation*}

By averaging and using $|S|\leq n-1$, there exists an $i$ such that
\begin{equation*}
|D_i|\geq \bigg(r + \frac{1}{r-1}\bigg)n -1 -\frac{|S|}{r-1}\geq rn -1 +\frac{1}{r-1}\,.
\end{equation*}
Since $|D_i|$ is an integer it is at least $rn$.
Furthermore if $|S|\leq n-r$ it is at least $rn+1$.
Therefore if $|S|\leq n-r$ we can find a suitable $D_i$.
If $|S|$ is larger than this, the next claim shows that either we still have a $D_i$ with $|D_i|\geq rn+1$ or we have at least two of size at least $rn$.
In the latter case, we will find one of these with $|T_i^\Delta|=0$.
\begin{claim}\label{cl:advave}
Suppose $|S|=n-\ell$ for some $1\leq \ell \leq r-1$.  Then either there exist $\ell$ values of $i$ for which
$|D_i| \geq rn$
or there exists one value of $i$ for which $|D_i| \geq rn +1$.
\end{claim}
\begin{claimproof}
Suppose for contradiction that the $\ell -1$ largest sets $D_i$ all have at most $rn$ vertices whilst all others have at most $rn-1$.
This gives
\[|S|+\sum_{i=1}^{r-1}|D_i| \leq n-\ell +(\ell-1)rn +(r-\ell)(rn-1)=(r^2-r+1)n-r <N\]
achieving a contradiction.
\end{claimproof}



Since $|T|\leq n-1 -|S|$, we see $|T| \leq \ell -1 $ and so if there are at least $\ell$ choices of $D_i$ with $|D_i|\geq rn$ then by the pigeon-hole principle at least one must have $|T_i^\Delta|=0$.


Our method now will essentially be to cover $S_i \cup T_i$ with disjoint copies of $K_r$ and then extend these to cover $B_i$ using that this is a blue clique and the total size of these sets is big enough.

We begin using only the assumption $|D_i|\geq rn$. 
Then \[ |B_i|\geq rn-|S_i|-|T_i^*|-|T_i^\Delta|\]
and so $|B_i|-(r-1)(|S_i|+|T_i^*|) \geq r(n-|S_i|-|T_i^*|)-|T_i^\Delta|\geq (\frac{r}{3}-1)|T_i^\Delta| +r \geq 1$.
The second inequality follows from $n-|S_i|-|T_i^*|\geq \frac13 |T_i^\Delta|+1$.

Since all vertices of $S_i \cup T_i^*$ are adjacent in blue to all but at most one vertex of $B_i$, we can extend all vertices of $S_i \cup T_i^*$ to disjoint blue copies of $K_r$ using $B_i$.
We now look to find $n-|S_i|-|T_i^*|$ disjoint blue copies $K_r$ on the remaining vertices of $B_i$ and $T_i^\Delta$.
Let $\tilde{B_i}$ denote the remaining vertices of $B_i$, noting that $|\tilde{B_i}| \geq (\frac{r}{3}-1)|T_i^\Delta|+r$.
Recall that edges from $T_i^\Delta$ to $B_i$ are blue and between different red triangles of $T_i^\Delta$ edges are blue.

If $|T_i^\Delta|=0$, then since $B_i$ is a blue clique we can find the remaining blue copies of $K_r$ entirely on the rest of $B_i$.

Remembering that $|T_i^\Delta|$ is a multiple of three, suppose $|T_i^\Delta|\geq 6$.  If $|T_i^\Delta|\geq 3r$, we first take blue copies of $K_r$ on $T_i^\Delta$ such that the vertices of $T_i^\Delta$ that are not covered by these blue cliques consist of $t$ red triangles with $2\leq t \leq r-1$.
We then cover the rest of $T_i^\Delta$ with blue copies of $K_r$ by greedily taking one vertex from each of the $t$ red triangles and extending this set of $t$ vertices to a blue $K_r$ using vertices from $\tilde{B_i}$.
In this process three blue copies of $K_r$ use both vertices of $T_i^\Delta$ and $\tilde{B_i}$, one for each vertex of the red triangles, and each such blue $K_r$ used $r-t$ vertices from $\tilde{B_i}$.
So long as $|\tilde{B_i}|\geq 3(r-t)$, this greedy procedure is successful.
Since we have $|\tilde{B_i}| \geq (\tfrac{r}{3}-1)|T_i^\Delta|+r$, we also have $|\tilde{B_i}|\geq 3t(\tfrac{r}{3}-1)+r$, which is at least $3(r-t)$ for $t \geq 2$.
If $t=1$ then it must be the case that $|T_i^\Delta| \geq 3(r+1)$ and so $|\tilde{B_i}|\geq (\tfrac{r}{3}-1)3(r+1)+r \geq 3(r-t)$.




Finally, suppose $|T_i^\Delta|=3$. Using Claim~\ref{cl:advave} we may assume $|D_i|\geq rn+1$. 
We have that $|\tilde{B_i}|\geq rn+1 -(r-1)(|S_i|+|T_i^*|) \geq \big(\tfrac{r}{3}-1\big)|T_i^\Delta| +r+1=2(r-1)$, and so we can extend two of the vertices of $T_i^\Delta$ to blue copies of $K_r$ using $\tilde{B_i}$. If necessary we then cover the rest of $\tilde{B_i}$ with more copies of $K_r$.
This completes the proof.
\end{proof}

\bibliographystyle{plain}




\end{document}
